Author's response to reviews

Title: A systematic review of musculoskeletal disorders among school teachers

Authors:

    Patience N Erick (patience.erick@uon.edu.au)
    Derek R Smith (Derek.Smith@newcastle.edu.au)

Version: 3 Date: 11 November 2011

Author's response to reviews:

The following revisions have been made as requested:

MAJOR REVISIONS:

Affiliations: Affiliations have been written as per the given format.
Authors' contributions: We have indicated that both authors have read and approved the final manuscript.
Additional files: Additional files have been removed from the system.
Tables: Figure 1 and Table 1 have been included in the manuscript. Figure 1 follows the Reference section while Table 1 follows Figures. Both Figures and tables have been removed from the submission system.

MINOR REVISIONS:

Structure: Heading 'Introduction' has been changed to 'Background.'
Typography: The authors have gone through the manuscript to check for typographical errors and make any final corrections as suggested.

Kind regards.